Author’s response to reviews

Title: Female Genital Cutting (FGC) and the Ethics of Care: the need for cultural competency and community engagement at the interface of health and migration

Authors:

Bilkis Vissandjée (bilkis.vissandjee@umontreal.ca)
Shereen Denetto (shereend@serc.mb.ca)
Paula Migliardi (paulam@serc.mb.ca)
Jodi Proctor (jodi.proctor@mail.mcgill.ca)

Version: 2 Date: 19 June 2013

Author's response to reviews:

A signed and dated letter would not upload adequately.
thanks